Author’s response to reviews

Title: When is Hastened Death Considered Suicide? A Systematically Conducted Literature Review about Palliative Care Professionals' Experiences Where Assisted Dying is Legal

Authors:

Sheri Gerson (sherimila@gmail.com)
Amanda Bingley (a.bingley@lancaster.ac.uk)
Nancy Preston (n.j.preston@lancaster.ac.uk)
Anne Grinyer (annegrinyer50@gmail.com)

Version: 2 Date: 25 Jul 2019

Author’s response to reviews:

Editor Comments:

1. Please add a "Consent for publication" section in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript.

Completed. Page 17, lines 22, 23

2. Please include a statement in Authors’ contributions indicating that all authors read and approved the final version of the manuscript.

Completed. Page 18, lines 8, 9

3. Please confirm the numbering of the tables, as we noticed references to Table 5 in the main text of the manuscript, but only four tables uploaded.

Amended. Page 8, line 8 and line 12 are corrected and now refer to Table 4. There is no Table 5.
4. In accordance with our submission guidelines, please add a “Results” heading in the main text of your manuscript.

Results heading has replaced heading called 'Themes' on page 8, line 10

5. As Nancy Preston is a member of the editorial board (Section Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Added statement in Competing Interests section on page 17, line 29

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.